Author's response to reviews

Title: A systematic review of the prevalence and determinants of nonadherence to phosphate binding medication in patients with end-stage renal disease

Authors:

Christina Karamanidou (psy77ck@yahoo.com)
Jane Clatworthy (jane.clatworthy@pharmacy.ac.uk)
John Weinman (john.weinman@kcl.ac.uk)
Rob Horne (rob.horne@pharmacy.ac.uk)

Version: 3 Date: 21 January 2008

Author's response to reviews: see over
Dr Sandra Lê  
BioMed Central Ltd  
Science Navigation Group  
Middlesex House  
34-42 Cleveland Street  
London W1T 4LB  

21st January 2008  

Dear Dr Lê  

RE: 1406841443156511 - A systematic review of the prevalence and determinants of nonadherence to phosphate binding medication in patients with end-stage renal disease  

Please find attached the revised review. We have addressed the following issues:  

- Please include the country at the end of the affiliation details.  
- Section headings: Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT  
- Competing interests - If there are none to declare, please write `The authors declare that they have no competing interests’  
- Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Please also ensure that each figure is a maximum size of one side of A4.  

There was one final point about the Authors’ contribution section:  

- Authors’ contributions - Please place the Authors’ contributions section towards the end of the manuscript text, directly preceding the Acknowledgements.  

This section was already situated towards the end of the manuscript before the Acknowledgement section. Please let me know if we need to make any changes to this section.  

If you require any further information, please do not hesitate to contact me at rob.horne@pharmacy.ac.uk or on 020 7874 1293.  

Yours sincerely  

Rob Horne  
Professor of Behavioural Medicine  
Director, Centre for Behavioural Medicine  

Department of Practice and Policy  
T + 44 (0) 207 874 1293  
F + 44 (0) 207 387 5693  
E rob.horne@pharmacy.ac.uk